

REVIEW HISTORY

Paper: **“Monetary Policy and Economic Growth in Nigeria: An Ardl-Bound Testing and ECM Approach”**

Corresponding Author: ADEREMI Timothy Ayomitunde

Email: aderemi.timothy@gmail.com

Doi: 10.19044/elj.v6no4a1

Peer review:

Reviewer 1: Blinded

Reviewer 2: Blinded

Published: 29.12.2019

